Author's response to reviews

Title: The symptom experience of people living with HIV and AIDS in the Eastern Cape, South Africa

Authors:

Karl K Peltzer (KPeltzer@hsrc.ac.za)
Nancy N Phaswana-Mafuya (NPhaswanaMafuya@hsrc.ac.za)

Version: 4 Date: 17 December 2008

Author's response to reviews:

the title, abstract, author details had been updated before, and are correct